Author's response to reviews

Title: Dual regulation of cell death by Akt kinase inhibitor MK-2206 in colorectal cancer

Authors:

Ekta Agarwal (ekta.agarwal@unmc.edu)
Anathbandhu Chaudhuri (achaudhuri@stillman.edu)
Premila D Leiphrakpam (pleiphrakpam@unmc.edu)
Katie L Haferbier (katie.haferbier@unmc.edu)
Michael G Brattain (mbrattain@unmc.edu)
Sanjib Chowdhury (schowdhury@unmc.edu)

Version: 2 Date: 2 July 2013

Author's response to reviews:

The following sentence was added to the Methods section under Xenograft Studies:

All the experiments involving animals were approved by the University of Nebraska Medical Center Institutional Animal Care and Use Committee.